Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL102744

Key Points:

- SNOWPACK's performance in calculating melt days shows agreement with the microwave observations in terms of total melt days and variability
- The relationship between cumulative melt days and modeled meltwater volume each melt season is strongly non-linear
- SNOWPACK-calculated annual melt days and meltwater production volume decrease at a small, but significant, rate from 1980 to 2021

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. F. Banwell. alison.banwell@colorado.edu

Citation:

Banwell, A. F., Wever, N., Dunmire, D., & Picard, G. (2023). Quantifying Antarctic-wide ice-shelf surface melt volume using microwave and firn model data: 1980 to 2021. Geophysical Research Letters, 50, e2023GL102744. https://doi. org/10.1029/2023GL102744

Received 4 JAN 2023 Accepted 19 MAY 2023

© 2023. The Authors. This is an open access article under

the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

ŗ **Quantifying Antarctic-Wide Ice-Shelf Surface Melt Volume**

Alison F. Banwell¹, Nander Wever^{2,3}, Devon Dunmire², and Ghislain Picard⁴

Using Microwave and Firn Model Data: 1980 to 2021

¹Cooperative Institute for Research in Environmental Sciences (CIRES), University of Colorado Boulder, Boulder, CO, USA, ²Department of Atmospheric and Oceanic Sciences (ATOC), University of Colorado Boulder, Boulder, CO, USA, ³WSL Institute for Snow and Avalanche Research SLF, Davos, Switzerland, ⁴Institut des Géosciences de l'Environnement (IGE), CNRS, University Grenoble Alpes, Grenoble, France

Abstract Antarctic ice-shelf stability is threatened by surface melt, which has been implicated in several ice-shelf collapse events over recent decades. Here, we first analyze cumulative days of wet snow/ice status ("melt days") for melt seasons from 1980 to 2021 over Antarctica's ice shelves using passive and active microwave satellite observations. As these observations do not directly reveal meltwater volumes, we calculate these using the physics-based multi-layer snow model SNOWPACK, driven by the global climate-reanalysis model Modern-Era Retrospective analysis for Research and Applications Version 2. We find a strong non-linear relationship between melt days and meltwater production volume. SNOWPACK's calculation of melt days shows agreement with observations of both cumulative days, and spatial and interannual variability. Highest melt rates are found on the Peninsula ice shelves, particularly in the 1992/1993 and 1994/1995 austral summers. Over all ice shelves, SNOWPACK calculates a small, but significant, decreasing trend in both annual melt days and meltwater production volume over the 41 years.

Plain Language Summary Antarctic ice shelves, which are floating glacier ice extending from the continent into the ocean, are important for the ice sheet's overall stability. However, they are threatened by surface melting, which has been linked to several ice-shelf collapse events in recent decades. Here, we analyze the duration and amount of surface melting on Antarctica's ice shelves from 1980 to 2021. We first analyze microwave satellite data to identify the number of "melt days" (days when meltwater is present). Second, as meltwater volumes cannot be quantified from microwave data, we calculate these using the snow model SNOWPACK. Our modeled melt days show good agreement with the satellite observations of both total days and the variability from year-to-year, giving us confidence in the model's ability to also simulate realistic temporal and spatial variations in meltwater volumes. We find a strong non-linear relationship between the number of melt days and meltwater volumes each summer. We also find that the highest meltwater volumes are produced on the Peninsula, reaching a peak in the 1992/1993 and 1994/1995 austral summers. Across all ice shelves, SNOWPACK calculates a small, but significant, decreasing trend in both annual melt days and meltwater production volume over the 41 years.

1. Introduction

About 75% of Antarctica is buttressed by floating ice shelves (Fürst et al., 2016), which regulate the rate that grounded ice is lost to the ocean (Bell et al., 2018). Since the 1990s, many ice shelves have thinned, and in some cases disintegrated (Banwell et al., 2013; Robel & Banwell, 2019; Scambos et al., 2003, 2009), due to both basal and surface melting (Adusumilli et al., 2020; Smith et al., 2020), resulting in increased ice flow from upstream grounded glaciers (Scambos et al., 2004).

Here we focus on surface and near-surface melt on ice shelves. When meltwater refreezes on ice shelves, the firm air content (FAC) decreases, which can lead to the formation of impermeable ice lenses, increasing the potential for future surface meltwater ponding (Bell et al., 2018; Kuipers Munneke et al., 2014). This process may threaten ice-shelf stability through increased potential for surface meltwater-induced flexure and hydrofracture initiation (Banwell & MacAyeal, 2015; Banwell et al., 2019; Scambos et al., 2009). With projected future atmospheric warming, models suggest that surface meltwater production on ice shelves will increase non-linearly due to the positive melt-albedo feedback (Gilbert & Kittel, 2021; Trusel et al., 2015). As a result, ice shelves are predicted to become more vulnerable to future surface meltwater-induced instability and breakup events (Lai et al., 2020).



Understanding current and predicting future Antarctic-wide ice-shelf vulnerability to surface meltwater-induced breakup requires a detailed, historical record of meltwater production. Spaceborne microwave sensors and climate models including a detailed firn representation are currently some of the primary tools for analyzing the presence of surface and near-surface melt in Antarctica (Banwell et al., 2021; Johnson et al., 2021; Kittel et al., 2018, 2021; Picard et al., 2022; van Wessem et al., 2018). However, the ability to quantify melt production from these tools is challenging because: (a) microwave sensors can only detect if meltwater is present or not, but they cannot directly quantify meltwater (Picard et al., 2007, 2022), and (b) climate models generally lack sufficient in-situ snow wetness and density data for validation of their firn components.

Here, we address the limitations outlined above by combining microwave radiometer (i.e., passive) and scatterometer (i.e., active) data with output from the physics-based, detailed, multi-layer snow model SNOWPACK, driven by the global climate reanalysis model Modern-Era Retrospective analysis for Research and Applications Version 2 (MERRA-2) (Gelaro et al., 2017). We first evaluate model performance by comparing annual cumulative melt days (i.e., days when meltwater is present) from SNOWPACK against those derived from the microwave data. Then, we analyze SNOWPACK-calculated meltwater production volume over all ice shelves from 1980 to 2021. We build upon a previous analysis of RACMO2 simulations for the period 1979 to 2010 by Kuipers Munneke et al. (2012), who found an insignificant decrease in surface meltwater production volume, by extending the analyzed time period to 2021, and using a more sophisticated firm model.

2. Data and Methodology

2.1. Ice Shelves and Ice Shelf Regions

We define ice-shelf boundaries (i.e., calving fronts, grounding lines and margins) using the MEaSUREs v2 data set (Mouginot et al., 2017), which was updated most recently in 2016. To maintain a constant area of the ice shelves through the 41 years of the study, we use these outlines for the entire period, thereby excluding any areas that may have either broken-up or grown in size between 1980 and 2016. For regional melt pattern analysis, we divide the ice shelves into eight regions following Alley et al. (2018) (Figure 1a).

2.2. Large-Scale Microwave Radiometer Observations of Melt

Microwave radiometers (19 GHz) record brightness temperatures that depend primarily on the snow temperature and emissivity (Colliander et al., 2022; Zwally, 1977). When liquid water exists in the snow, absorption significantly increases, resulting in increased microwave emissivity, and thus substantially higher brightness temperatures (Liu et al., 2006; Zwally & Fiegles, 1994). Here, we used the near-daily polar stereographic 25 km melt product (version 2) of Torinesi et al. (2003) and Picard and Fily (2006) updated for austral summers from 1980 to 2021. This melt product, widely used before (e.g., Banwell et al., 2021; Magand et al., 2008; Wille et al., 2019), provides binary wet/dry status using observations from five successive sensors from 1980 to 2021 (Scanning Multi-channel Microwave Radiometer and a series of Special Sensor Microwave Imager (SSM/I) sensors). Based on radiative transfer simulations, radiometer brightness temperatures are typically sensitive to the dry/wet status in the uppermost ~2 m of snow/ice (Picard et al., 2022).

We used data from melt seasons (November 1 to March 31 inclusive) from 1980/1981 to 2020/2021, apart from 1987/1988 due to 41 days of missing data. For each melt season and each 25 km grid cell within the ice-shelf mask, we calculated a daily time series of radiometer-derived wet/dry status, resulting in a calculation of cumulative "melt days". Throughout this paper, we use the phrase "melt days" when referring to days of "wet status" in the microwave data, but note that active melting may not be occurring; meltwater may only be present, and/or may be in the process of refreezing.

2.3. Small-Scale Microwave Scatterometer Observations of Melt

We also derived smaller-scale (4.45 km) melt information from C-band (5.225 GHz) radar backscatter images collected by EUMETSAT's Advanced SCATterometer (ASCAT). The 4.45 km enhanced-resolution product was obtained by applying the Scatterometer Image Reconstruction (SIR) algorithm with filtering (Lindsley & Long, 2016) to improve the spatial resolution (Early & Long, 2001). For each day and grid cell, meltwater is assumed to be present when the ASCAT signal is lower than the winter mean signal minus 3 dB (Ashcraft



Figure 1. Observed and modeled melt days. (a) Ice shelves divided into eight regions (Alley et al., 2018). For individual ice-shelf locations, see Figure S1 in Supporting Information S1. (b) Passive versus active microwave-derived melt days for each melt season from 2007/2008 to 2019/2020. (c) Passive microwave-derived melt days versus SNOWPACK melt days for each melt season from 1980/1981 to 2020/2021. (d) Active microwave-derived melt days versus SNOWPACK melt days for each melt season from 2007/2008 to 2019/2020. The dashed black lines show y = x. The solid black lines are the regression lines. All R² values are significant at p < 0.05. Scatter points are colored according to the ice shelf regions (panel a).

& Long, 2006; Banwell et al., 2021) In locations where snow and firn layers are devoid of liquid meltwater, ASCATS's penetration depth is of the order of meters to tens of meters, hence ASCAT is more likely to be sensitive to water present at depth than microwave radiometers (Picard et al., 2022).

For melt seasons from 2007/2008 to 2019/2020, and for each grid cell within the ice-shelf mask, we calculated the daily time series of scatterometer-derived wet/dry snow status, and hence cumulative melt days.

2.4. Firn Model SNOWPACK

To calculate days when meltwater is present in the firn, meltwater production volume, and the presence of surface meltwater versus meltwater at depth, we use the detailed, physics-based, multi-layer model SNOWPACK (Bartelt & Lehning, 2002; Lehning et al., 2002a, 2002b). SNOWPACK has been extensively validated for liquid water flow processes (e.g., Wever et al., 2014, 2015, 2016), and has been used in polar regions including Antarctica (Dunmire et al., 2020, 2021; Keenan et al., 2021; Steger et al., 2017; Thompson-Munson et al., 2023).

SNOWPACK was forced with hourly air temperature, relative humidity, wind speed, incoming short- and longwave radiation, and precipitation from MERRA-2 (Gelaro et al., 2017). Gossart et al. (2019) report generally good agreement between MERRA-2 and observed near-surface climatology, including 2 m air temperature and wind speeds, at elevations <500 m in Antarctica, including the ice shelves.

This study analyzes melt in all austral summers from 1980/1981, the first complete summer when MERRA-2 data are available, to 2020/2021. At the \pm 70° latitude bands, MERRA-2 has a resolution of 24 km × 56 km. To minimize model run time, MERRA-2 points (i.e., grid cell centers) were subsampled based on the similarity in the spatial distribution of climatological conditions following Smith et al. (2020), whereby points were retained in areas of high spatially varying climatic conditions, but removed where climatic conditions were similar to neighboring points. From this sub-selection of MERRA-2 points, we ran SNOWPACK at all points that were located both: (a) on ice shelves >625 km² (i.e., at least one 25 km passive microwave grid cell); and (b) <200 m elevation above local sea level according to MERRA-2's model topography. In total, there were 847 MERRA-2 points that fulfilled both criteria.

Using the model setup described by Keenan et al. (2021) and the spin-up procedure from Thompson-Munson et al. (2023), SNOWPACK was first repeatedly run at each of these 847 MERRA-2 points over the period 1980 to 2021 until the spin-ups reached either (a) a >150 m thick column of snow, firn, and ice, or (b) the bottom 2 m of the simulated firn column consisted of solid ice with a total firn column >10 m depth. At 11 grid points, neither of these conditions were reached due to high ablation, restricting the final analysis to 836 grid points, located on 36 ice shelves (Figure S1 in Supporting Information S1). As the ice-shelf floats on ocean water, we prescribed a fixed lower-boundary temperature of -1.8° C; the freezing point of ocean water. This deviates from Keenan et al. (2021), where the annual average air temperature was prescribed for this lower boundary. While that approach is adequate for dry, grounded ice, it no longer holds for floating ice with surface melt.

SNOWPACK calculates meltwater production via a physically based energy-balance model that captures processes such as changes in accumulation, snow-albedo feedback, percolation and latent heat release through the entire firn column (e.g., Wever et al., 2014, 2015, 2016). We expect this to better capture complex melt dynamics compared to, for example, the positive-degree day parameterization using air temperature only, as used in the Community Firn Model (e.g., Medley et al., 2022; Vandecrux et al., 2020). To calculate meltwater volumes over all areas within the ice-shelf mask, we use the method of radial basis functions to interpolate SNOWPACK-simulated melt at each of the 836 MERRA-2 points.

2.5. SNOWPACK Validation Against Microwave Data

To evaluate SNOWPACK's performance in calculating realistic spatial and temporal variations in melt, we compared modeled and microwave-derived cumulative melt days for each melt season from 1980/1981 to 2020/2021 for the passive microwave data, and from 2007/2008 to 2019/2020 for the active microwave data, at every one of the 836 MERRA-2 forcing grid points. For each point, we compared the closest radiometer and scatterometer ice-shelf grid cells.

From the SNOWPACK output, we calculated cumulative melt days over different ranges of snow depths from the ice surface (top most 0.1 m, to the complete snowpack depth), and for different thresholds of minimum liquid water content (0–2 mm). We then calculated the root mean square error (RMSE) between the observed and modeled number of melt days, for each of the 836 MERRA-2 points over the 41 melt seasons for the passive data and 13 melt seasons for the active data.

As acquisitions of the radiometers and scatterometers are in the local mornings and evenings (Lindsley & Long, 2016; Picard & Fily, 2006), mean statistics from SNOWPACK were calculated from 6 to 10 a.m. and 6 to 10 p.m. local time (defined by grid point longitude), to more adequately compare to the time-constrained melt observations.

3. Results and Discussion

3.1. Evaluation of Microwave-Derived and SNOWPACK Calculated Melt Days

We first compare cumulative melt days in each melt season from 1981 to 2020 observed by passive versus active microwave data at each of the 836 MERRA-2 grid point locations (Figure 1b, RMSE = 6.8 days). Our results





Figure 2. Passive microwave-derived and modeled melt days. (a–g) Passive microwave-derived versus SNOWPACK melt days for each melt season from 1980/1981 to 2020/2021 for all selected Modern-Era Retrospective analysis for Research and Applications Version 2 (MERRA-2) points in 7 of the 8 regions (Victoria Land is excluded as it only contains one MERRA-2 point). (h) Mean values (colored dots) for each 7 region, with the whiskers representing \pm one standard deviation. Dashed black lines show y = x, and solid black lines are the regression lines. All R^2 values are significant at p < 0.05. Note the different scales on the axis. Plot colors correspond to the region colors (Figure 1a).

suggest that for areas of low cumulative melt days (i.e., <40), for example, in the Ronne-Filchner and Ross regions (Figure 1b, Figures S2e and S2f in Supporting Information S1), active microwave observes fewer melt days compared to passive microwave. But for areas with more melt days (i.e., >80), for example, the Peninsula (Figure 1b and Figure S2d in Supporting Information S1) and in particular Wilkins, where microwave-derived melt days are higher than for any other ice shelf (Figure S3 in Supporting Information S1), the active microwave data reports more melt days than the passive data. These findings make sense, as active microwave is less sensitive to weak surface melt than passive microwave, but active microwave can detect meltwater deeper in the snow-pack due to its longer wavelength, and so it often detects meltwater much later into the fall (Picard et al., 2022; Weber Hoen & Zebker, 2020).

Second, we compare cumulative melt days derived from passive and active microwave to those calculated by SNOWPACK over a range of firn depths, and for a range of minimum meltwater detection thresholds (Table S1 in Supporting Information S1) as described in Section 2.5. For both the passive and active microwave data, the SNOWPACK output calculated from the surface to a 0.1 m firn depth and a minimum meltwater threshold of 0 mm produces the lowest mean RMSEs (8.2 and 10.9 days for passive and active, respectively). Using this optimum SNOWPACK output, the bias between the modeled and passive microwave-derived melt days, for all MERRA-2 points and for all melt seasons, is -0.14 melt days (Figure 1c), indicating that SNOWPACK slightly underestimates melt days. Considering the ice-shelf regions separately (Figures 2a-2g), the agreement between modeled and passive microwave melt days is less good. The lowest biases and RMSE values between these data are found in the Ronne-Filchner and Ross Sea regions, despite high scatter there. When just mean melt days in the seven regions are compared, the agreement is again, better (Figure 2h).

Using this same optimum SNOWPACK output, comparing modeled and active microwave-derived melt days for all MERRA-2 points from 2007/2008 to 2021/2022 indicates that, again, SNOWPACK slightly underestimates melt days (Figure 1d; bias = 0.51 melt days). At a regional level (Figures S4a–S4g in Supporting Information S1), the agreement is, again, less good. However, when the mean melt days in the seven regions are compared, the agreement is again, better (Figure S4h in Supporting Information S1).



Geophysical Research Letters



Figure 3. Timeseries of observed and modeled melt days. (a) Microwave and modeled cumulative melt days for each melt season from 1980/1981 to 2020/2021 (from 2007/2008 to 2020/2021 for active microwave) for all regions, and (b–h), for individual regions. The numbers next to the region names indicate the total Modern-Era Retrospective analysis for Research and Applications Version 2 points in each region. Note the different *y*-axis scales. Dates on the *x*-axes indicate the second year of each austral summer, for example, 2020 corresponds to 2019/2020. The standard deviation on each plot is the interannual variability of the passive microwave data, and root mean square error and R^2 are calculated between passive microwave and SNOWPACK data. All R^2 values are significant at p < 0.05.

Although the precise number of microwave-derived and modeled cumulative melt days differs slightly between the datasets in each region, the patterns of interannual variability are relatively consistent between both microwave sensors and SNOWPACK (Figure 3). Considering all 836 points in all regions (Figure 3a), the RMSE is 68% of the standard deviation, suggesting that the model has some capability in reproducing the observed interannual variability in melt, and also that the climatic conditions are generally well represented in the MERRA-2 reanalysis used to force SNOWPACK.

Within individual regions, we observe local-scale spatial variations in the agreement between passive microwave-derived and modeled melt days (Figure S5 in Supporting Information S1). For example, SNOW-PACK underestimates melt days in the southwest Peninsula, particularly on the Wilkins and south George VI (bias <-30 days), whereas it overestimates melt days on the Scar Inlet and north Larsen C (bias <15 days). These spatial variations may be partially attributable to the relatively coarse grid of MERRA-2, meaning that SNOWPACK fails to capture localized atmospheric processes (see Section 3.2). Lateral meltwater flow (e.g., Dell et al., 2020) may also contribute to discrepancies.

3.2. Modeled Meltwater Volume on Ice Shelves, 1980 to 2021

Over the 41-year time period from 1980/1981 to 2020/2021, modeled mean meltwater volume over all ice shelves is 43.2 Gt yr⁻¹ with high interannual variability (std. dev. = 18.3 Gt yr⁻¹) (Figure 4a). Record high and low meltwater volumes were produced in 1997/1998 (88.1 Gt) and 1985/1986 (16.5 Gt), respectively.

SNOWPACK-calculated mean meltwater volumes over all ice shelves from 1980/1981 to 2020/2021 are lower than the equivalent values from two polar regional climate models, modèle atmosphérique régional (MAR) v3.12 (Kittel et al., 2021) and the regional atmospheric model (RACMO) RACMO2.3p2 (van Wessem et al., 2018, 2023); 74.5 Gt yr⁻¹ and 68.1 Gt yr⁻¹ respectively. However, when we compare SNOWPACK melt (mm w.e.) simulated at the 836 MERRA-2 points to melt calculated in the nearest MAR and RACMO grid cells, for each region and over the 41-year time period, we show that SNOWPACK's melt underestimation is less substantial (Figure S6 in Supporting Information S1). For MAR versus SNOWPACK, the regression line ($R^2 = 0.98$) slope is 0.66, and for RACMO versus SNOWPACK, the regression line ($R^2 = 0.94$) slope is 0.93. In





Figure 4.

comparison, for RACMO versus MAR, the regression line ($R^2 = 0.94$) slope is 1.39, indicating that at these 836 points, modeled melt from the three models is comparable.

As mentioned earlier, differences in the spatial resolution between the higher-resolution models RACMO and MAR and the lower-resolution product MERRA-2 may explain why SNOWPACK melt rates are generally slightly lower. MERRA-2's coarse grid will not be able to capture localized atmospheric processes that often act to increase local melt rates. Such processes include periodic foehn winds over Larsen C (Elvidge et al., 2016), and katabatic winds on Roi Baudouin (Dunmire et al., 2020; Lenaerts et al., 2017) that blow away snow revealing blue ice. For north George VI, the MERRA-2 climate simply cannot be resolved due to high topography on either side of this narrow ice shelf. Therefore we cannot run SNOWPACK here, so no melt is modeled here. As we interpolate point-simulated melt over all ice-shelf areas to calculate volumes (Figure S7 in Supporting Information S1), melt volumes over north George VI are therefore underestimated compared to the high melt observed here (e.g., Banwell et al., 2021; Holt et al., 2013).

We find a significant (p < 0.05) decreasing trend in SNOWPACK-calculated annual meltwater production volume of -0.55 Gt yr⁻² from 1980/1981 to 2020/2021 (Figure 4a). This finding is in line with significant (p < 0.05) decreasing trends in modeled and passive microwave-derived annual melt days (-0.078 and -0.061 d yr⁻², respectively) over the same time period. Additionally, the MERRA-2 mean summer air temperature trend over all ice shelves and over the same time period is negative (-0.016 C yr⁻²), however it is only significant at p < 0.1(Figure 4a). RACMO2.3 and MAR v3.12 melt volumes over all ice shelves and the same time period (not shown) also show decreasing, albeit not significant, trends of -0.27 and -0.09 Gt yr⁻² respectively.

Considering meltwater production volumes in the eight ice-shelf regions (Figure 4b), by far most of Antarctica's meltwater is produced on the Peninsula (mean = 24.6 Gt yr⁻¹) where mean melt rates are <435 mm w.e. yr⁻¹ on the Scar Inlet and north Larsen C Ice Shelf (Figure S7 in Supporting Information S1). After the Peninsula, the next highest meltwater volumes are produced in Wilkes Land (mean 4.5 Gt yr⁻¹), the Amery Region (mean 4.4 Gt yr⁻¹), and Dronning Maud Land (mean 4.2 Gt yr⁻¹) (Table S2 in Supporting Information S1). Ronne-Filchner, Ross Sea, and Victoria Land all have mean melt volumes <2 Gt yr⁻¹. Over the 41-year period, all eight regions experience overall decreasing meltwater production rates (Table S2 in Supporting Information S1), though these trends are only significant at p < 0.05 for the Amery, Amundsen Sea, and Dronning Maud Land regions (Wilkes Land is significant at p < 0.1). Notably, the Peninsula's annual meltwater production volume reaches a maximum of 55.6 Gt in 1992/1993 before decreasing at a statistically significant rate (-0.88 Gt yr⁻², p < 0.05) until 2020/2021 (Figure 4b and Figure S7 in Supporting Information S1). We also see a significant decrease in MERRA-2 mean summer air temperature over the same time period (-0.06°C yr⁻², p < 0.05; Figure S8 in Supporting Information S1).

We find a strong non-linear positive relationship between melt days and meltwater production volume, varying between 0.1 Gt melt day⁻¹ for areas with meltwater production <5 Gt yr⁻¹ (e.g., Ross Sea, Ronne-Filchner) to 1.5 Gt melt day⁻¹ (Figure 4c) for areas with meltwater production >40 Gt yr⁻¹ (e.g., Peninsula in high melt seasons). This finding indicates how microwave data alone, which enables the quantification of "melt days," is not a good indicator of melt volumes. This is because, in warmer summers (Figure 4d), the melting point (0 C) is passed more frequently, the positive temperature-albedo feedback mechanism (Trusel et al., 2015) results in further melting, and larger volumes of meltwater will take longer to refreeze than smaller volumes. Hence we argue for the use of sophisticated energy-balance models such as SNOWPACK.

4. Conclusions

This study first provided a historical record of cumulative melt days over Antarctica's ice shelves for melt seasons from 1980 to 2021 using passive and active microwave satellite observations. Second, as these microwave observations do not reveal meltwater volumes, these were calculated using the snow model SNOWPACK, driven by MERRA-2. We found a strong non-linear positive relationship between melt days and meltwater production

Figure 4. Modeled meltwater production volume. (a) Modeled annual meltwater volume (Gt yr⁻¹) over all ice shelves (black line, left *y*-axis) and Modern-Era Retrospective analysis for Research and Applications Version 2 mean summer (November 1 to March 31) air temperature over all ice shelves (red solid line, right *y*-axis). The black and red dashed lines are the first order polynomials that best fit the solid black (p < 0.05) and solid red (p < 0.1) lines. (b) As in panel a, but partitioned into the 8 regions. The *x*-axes dates indicate the second year of each austral summer, for example, 2020 corresponds to 2019/2020 (1 April 2019 to 31 March 2020). (c) Passive microwave-derived melt days versus meltwater volumes for each region and melt season from 1980/1981 to 2020/2021. (d) Mean summer air temperature versus meltwater production volumes for each region and each year.

volume. Evaluation of SNOWPACK's performance in calculating melt days showed agreement with observations in terms of both cumulative days and variability, which gave us confidence in SNOWPACK's ability to also simulate realistic temporal and spatial relative variations in meltwater volumes.

Over the 41-year historical period, SNOWPACK-calculated meltwater volumes were highest on the Peninsula. Over all ice shelves, SNOWPACK showed small, but statistically significant (p < 0.05), decreasing trends in both annual melt days and meltwater production over this time period, in line with a significant decreasing, albeit not significant, trends in ice-shelf annual melt days. RACMO2.3 and MAR melt data also showed decreasing, albeit not significant, trends in ice-shelf annual meltwater production over this same time period. As refreezing of meltwater in firn decreases FAC, thereby increasing the potential for surface meltwater production over the past four decades found in this study may have helped to mitigate against ice-shelf instability. Finally, we note that despite the small changes in ice-shelf surface melt to date, projected atmospheric warming means that surface meltwater production on ice shelves is expected to increase non-linearly in the future, and hence ice shelves are predicted to become more vulnerable to future surface meltwater-induced instability (Kittel et al., 2021; Lai et al., 2020). As an example, Gilbert and Kittel (2021) have predicted that 34% of all Antarctic ice shelf areas may be vulnerable to collapse at 4°C of warming above pre-industrial levels.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

MERRA-2 data can be accessed online at https://gmao.gsfc.nasa.gov/reanalysis/MERRA-2/data_access/. The MEaSUREs v2 ice shelf outlines are available at https://doi.org/10.5067/AXE4121732AD (Mouginot et al., 2017). The passive microwave melt product is available at https://snow.univ-grenoble-alpes.fr/melting/. The ASCAT enhanced resolution product is available at https://www.scp.byu.edu/data/Ascat/SIR/msfa/Ant. html. The processed passive microwave melt data used in this study are available via https://doi.org/10.18709/ PERSCIDO.2022.09.DS376 (repository at https://perscido.univ-grenoble-alpes.fr/datasets/DS376), and the processed active microwave data are available via https://doi.org/10.18709/PERSCIDO.2023.05.DS394 (repository at https://perscido.univ-grenoble-alpes.fr/datasets/DS394). The SNOWPACK model is published under a GNU LGPLv3 license by the WSL Institute for Snow and Avalanche Research SLF at https://snowpack.slf.ch. The SNOWPACK model is published under a GNU LGPLv3 license by the WSL Institute for Snow and Avalanche Research SLF at https://snowpack.slf.ch. The repository used to develop the model code used in this study can be accessed via https://doi.org/10.5281/zenodo.3891845 (repository at: https://github.com/snowpack-model/snowpack). The exact model source code used in this study corresponds to commit 7dd9f58, and can be accessed via https://doi.org/10.5281/zenodo.7956900. The output from the SNOWPACK simulations, including the analyzed statistics calculated from the model output, can be accessed via https://doi.org/10.5281/zenodo.7956517. The code used to calculate the statistics from the SNOWPACK output files can be accessed via https://doi.org/10.5281/ zenodo.7957639 (repository at: https://github.com/nwever/SNOWPACK_iceshelves_melt).

References

Adusumilli, S., Fricker, H. A., Medley, B., Padman, L., & Siegfried, M. (2020). Interannual variations in meltwater input to the Southern Ocean from Antarctic ice shelves. *Nature Geoscience*, 13(9), 616–620. https://doi.org/10.1038/s41561-020-0616-z

Alley, K. E., Scambos, T. A., Miller, J. Z., Long, D. G., & MacFerrin, M. (2018). Quantifying vulnerability of Antarctic ice shelves to hydrofracture using microwave scattering properties. *Remote Sensing of Environment*, 210, 297–306. https://doi.org/10.1016/j.rse.2018.03.025

Ashcraft, I. S., & Long, D. G. (2006). Comparison of methods for melt detection over Greenland using active and passive microwave measurements. *International Journal of Remote Sensing*, 27(12), 2469–2488. https://doi.org/10.1080/01431160500534465

Banwell, A. F., Datta, R. T., Dell, R. L., Moussavi, M., Brucker, L., Picard, G., et al. (2021). The 32-year record-high surface melt in 2019/2020 on the northern George VI Ice Shelf, Antarctic Peninsula. *The Cryosphere*, 15(2), 909–925. https://doi.org/10.5194/tc-15-909-2021

Banwell, A. F., & MacAyeal, D. R. (2015). Ice-shelf fracture due to viscoelastic flexure stress induced by fill/drain cycles of supraglacial lakes. *Antarctic Science*, 27(6), 587–597. https://doi.org/10.1017/S0954102015000292

Banwell, A. F., MacAyeal, D. R., & Sergienko, O. V. (2013). Breakup of the Larsen B Ice Shelf triggered by chain reaction drainage of supraglacial lakes. *Geophysical Research Letters*, 40(22), 5872–5876. https://doi.org/10.1002/2013GL057694

Banwell, A. F., Willis, I. C., Macdonald, G. J., Goodsell, B., & MacAyeal, D. R. (2019). Direct measurements of ice shelf flexure caused by surface meltwater ponding and drainage. *Nature Communications*, 10(1), 730. https://doi.org/10.1038/s41467-019-08522-5

Acknowledgments

AFB received support from the U.S. National Science Foundation under Award #1841607 to the University of Colorado Boulder, and from a University of Colorado Boulder Cooperative Institute for Research in Environmental Studies (CIRES) Innovative Research Proposal (IRP) award. NW was supported by the National Aeronautics and Space Administration (NASA) under Award #80NSSC20K0969 issued by the "Studies with ICESat-2" program. DD was supported by a NASA FINESST Fellowship (Award #80NSSC19K1329). GP received support from the ESA Project ESA/AO/1-9570/18/I-DT----- "4D Antarctica." The authors thank Ludovic Brucker, Ted Scambos, Megan Thompson Munson and Rebecca Dell for useful discussions. The authors thank the Editor, Mathieu Morlighem, and two reviewers: Peter Kuipers Munneke and one anonymous reviewer, for their very useful comments.

- Bartelt, P., & Lehning, M. (2002). A physical SNOWPACK model for the Swiss avalanche warning: Part I: Numerical model. *Cold Regions Science and Technology*, *35*(3), 123–145. https://doi.org/10.1016/S0165-232X(02)00074-5
- Bell, R. E., Banwell, A. F., Trusel, L. D., & Kingslake, J. (2018). Antarctic surface hydrology and impacts on ice-sheet mass balance. Nature Climate Change, 8(12), 1044–1052. https://doi.org/10.1038/s41558-018-0326-3
- Colliander, A., Mousavi, M., Marshall, S., Samimi, S., Kimball, J. S., Miller, J. Z., et al. (2022). Ice sheet surface and subsurface melt water discrimination using multi-frequency microwave radiometry. *Geophysical Research Letters*, 49(4), e2021GL096599. https://doi. org/10.1029/2021GL096599

Dell, R., Arnold, N., Willis, I., Banwell, A., Williamson, A., Pritchard, H., & Orr, A. (2020). Lateral meltwater transfer across an Antarctic ice shelf. *The Cryosphere*, 14(7), 2313–2330. https://doi.org/10.5194/tc-14-2313-2020

- Dunmire, D., Banwell, A. F., Wever, N., Lenaerts, J. T. M., & Datta, R. T. (2021). Contrasting regional variability of buried meltwater extent over 2 years across the Greenland Ice Sheet. *The Cryosphere*, *15*(6), 2983–3005. https://doi.org/10.5194/tc-15-2983-2021
- Dunmire, D., Lenaerts, J. T. M., Banwell, A. F., Wever, N., Shragge, J., Lhermitte, S., et al. (2020). Observations of buried lake drainage on the Antarctic Ice Sheet. *Geophysical Research Letters*, 47(15), e2020GL087970. https://doi.org/10.1029/2020GL087970
- Early, D. S., & Long, D. G. (2001). Image reconstruction and enhanced resolution imaging from irregular samples. *IEEE Transactions on Geoscience and Remote Sensing*, 39(2), 291–302. https://doi.org/10.1109/36.905237
- Elvidge, A. D., Renfrew, I. A., King, J. C., Orr, A., & Lachlan-Cope, T. A. (2016). Foehn warming distributions in nonlinear and linear flow regimes: A focus on the Antarctic Peninsula. *Quarterly Journal of the Royal Meteorological Society*, 142(695), 618–631. https://doi. org/10.1002/qj.2489
- Fürst, J. J., Durand, G., Gillet-chaulet, F., Tavard, L., Rankl, M., Braun, M., & Gagliardini, O. (2016). The safety band of Antarctic ice shelves. *Nature Climate Change*, 6, 2014–2017. https://doi.org/10.1038/NCLIMATE2912
- Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., et al. (2017). The modern-era retrospective analysis for research and applications, version 2 (MERRA-2). Journal of Climate, 30(14), 5419–5454. https://doi.org/10.1175/JCLI-D-16-0758.1
- Gilbert, E., & Kittel, C. (2021). Surface melt and runoff on Antarctic ice shelves at 1.5°C, 2°C, and 4°C of future warming. *Geophysical Research Letters*, 48(8), e2020GL091733. https://doi.org/10.1029/2020GL091733
- Gossart, A., Helsen, S., Lenaerts, J. T. M., Broucke, S. V., van Lipzig, N. P. M., & Souverijns, N. (2019). An evaluation of surface climatology in state-of-the-art reanalyses over the Antarctic ice sheet. *Journal of Climate*, 32(20), 6899–6915. https://doi.org/10.1175/JCLI-D-19-0030.1
- Holt, T. O., Glasser, N. F., Quincey, D. J., & Siegfried, M. R. (2013). Speedup and fracturing of George VI Ice Shelf, Antarctic Peninsula. *The Cryosphere*, 7(3), 797–816. https://doi.org/10.5194/tc-7-797-2013
- Johnson, A., Hock, R., & Fahnestock, M. (2021). Spatial variability and regional trends of Antarctic ice shelf surface melt duration over 1979– 2020 derived from passive microwave data. *Journal of Glaciology*, 68(269), 1–14. https://doi.org/10.1017/jog.2021.112
- Keenan, E., Wever, N., Dattler, M., Lenaerts, J. T. M., Medley, B., Kuipers Munneke, P., & Reijmer, C. (2021). Physics-based SNOWPACK model improves representation of near-surface Antarctic snow and firn density. *The Cryosphere*, 15(2), 1065–1085. https://doi.org/10.5194/ tc-15-1065-2021
- Kittel, C., Amory, C., Agosta, C., Delhasse, A., Doutreloup, S., Huot, P.-V., et al. (2018). Sensitivity of the current Antarctic surface mass balance to sea surface conditions using MAR. *The Cryosphere*, 12, 3827–3839. https://doi.org/10.5194/tc-12-3827-2018
- Kittel, C., Amory, C., Agosta, C., Jourdain, N. C., Hofer, S., Delhasse, A., et al. (2021). Diverging future surface mass balance between the Antarctic ice shelves and grounded ice sheet. *The Cryosphere*, 15(3), 1215–1236. https://doi.org/10.5194/tc-15-1215-2021
- Kuipers Munneke, P., Ligtenberg, S. R. M., van Den Broeke, M. R., & Vaughan, D. G. (2014). Firn air depletion as a precursor of Antarctic ice-shelf collapse. *Journal of Glaciology*, 60(220), 205–214. https://doi.org/10.3189/2014jog13j183
- Kuipers Munneke, P., Picard, G., van den Broeke, M. R., Lenaerts, J. T. M., & van Meijgaard, E. (2012). Insignificant change in Antarctic snowmelt volume since 1979. Geophysical Research Letters, 39, L01501. https://doi.org/10.1029/2011GL050207
- Lai, C. Y., Kingslake, J., Wearing, M. G., Chen, P. C., Gentine, P., Li, H., et al. (2020). Vulnerability of Antarctica's ice shelves to meltwater-driven fracture. *Nature*, 584(7822), 574–578. https://doi.org/10.1038/s41586-020-2627-8
- Lehning, M., Bartelt, P., Brown, B., Fierz, C., & Satyawali, P. (2002a). A physical SNOWPACK model for the Swiss avalanche warning: Part II: Snow microstructure. *Cold Regions Science and Technology*, 35(3), 147–167. https://doi.org/10.1016/s0165-232x(02)00073-3
- Lehning, M., Bartelt, P., Brown, B., Fierz, C., & Satyawali, P. (2002b). A physical SNOWPACK model for the Swiss avalanche warning: Part III: Meteorological forcing, thin layer formation and evaluation. *Cold Regions Science and Technology*, 35(3), 169–184. https://doi.org/10.1016/ S0165-232X(02)00073-3
- Lenaerts, J. T. M., Lhermitte, S., Drews, R., Ligtenberg, S. R. M., Berger, S., Helm, V., et al. (2017). Meltwater produced by wind-albedo interaction stored in an East Antarctic ice shelf. *Nature Climate Change*, 7(1), 58–63. https://doi.org/10.1038/NCLIMATE3180
- Lindsley, R. D., & Long, D. G. (2016). Enhanced-resolution reconstruction of ASCAT backscatter measurements. *IEEE Transactions on Geosci*ence and Remote Sensing, 54(5), 2589–2601. https://doi.org/10.1109/TGRS.2015.2503762
- Liu, H., Wang, L., & Jezek, K. C. (2006). Spatiotemporal variations of snowmelt in Antarctica derived from satellite scanning multichannel microwave radiometer and Special Sensor Microwave Imager data (1978–2004). Journal of Geophysical Research, 111(F1), F01003. https:// doi.org/10.1029/2005JF000318
- Magand, O., Picard, G., Brucker, L., Fily, M., & Gen- thon, C. (2008). Snow melting bias in microwave mapping of Antarctic snow accumulation. *The Cryosphere*, 2, 109–115. https://doi.org/10.5194/tc-2-109-2008
- Medley, B., Neumann, T. A., Zwally, H. J., Smith, B. E., & Stevens, C. M. (2022). Simulations of firn processes over the Greenland and Antarctic ice sheets: 1980–2021. The Cryosphere, 16(10), 3971–4011. https://doi.org/10.5194/tc-16-3971-2022
- Mouginot, J., Scheuchl, B., & Rignot, E. (2017). MEaSUREs Antarctic boundaries for IPY 2007-2009 from satellite radar, version 2. Boulder, Colorado USA. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/AXE4121732AD
- Picard, G., & Fily, M. (2006). Surface melting observations in Antarctica by microwave radiometers: Correcting 26-year time series from changes in acquisition hours. *Remote Sensing of the Environment*, 104(3), 325–336. https://doi.org/10.1016/j.rse.2006.05.010
- Picard, G., Fily, M., & Gallee, H. (2007). Surface melting derived from microwave radiometers: A climatic indicator in Antarctica. Annals of Glaciology, 46, 29–34. https://doi.org/10.3189/172756407782871684
- Picard, G., Leduc-Leballeur, M., Banwell, A. F., Brucker, L., & Macelloni, G. (2022). The sensitivity of satellite microwave observations to liquid water in the Antarctic snowpack. *The Cryosphere*, 16(12), 5061–5083. https://doi.org/10.5194/tc-16-5061-2022
- Robel, A. A., & Banwell, A. F. (2019). A speed limit on ice shelf collapse through hydrofracture. *Geophysical Research Letters*, 46(21), 12092–12100. https://doi.org/10.1029/2019g1084397
- Scambos, T., Fricker, H. A., Liu, C. C., Bohlander, J., Fastook, J., Sargent, A., et al. (2009). Ice shelf disintegration by plate bending and hydrofracture: Satellite observations and model results of the 2008 Wilkins ice shelf break-ups. *Earth and Planetary Science Letters*, 280(1–4), 51–60. https://doi.org/10.1016/j.epsl.2008.12.027

- Scambos, T. A., Bohlander, J. A., Shuman, C. A., & Skvarca, P. (2004). Glacier acceleration and thinning after ice shelf collapse in the Larsen B embayment, Antarctica. *Geophysical Research Letters*, *31*(18), L18402. https://doi.org/10.1029/2004GL020670
- Scambos, T. A., Hulbe, C., & Fahnestock, M. A. (2003). Climate-induced ice shelf disintegration in the Antarctic Peninsula. In E. W. Domack, et al. (Eds.), Antarctic Peninsula climate variability: A historical and paleoenvironmental perspective, Antarctic Research series (Vol. 79, pp. 79–92). American Geophysical Union. https://doi.org/10.1029/AR079p0079
- Smith, B., Fricker, H. A., Gardner, A. S., Medley, B., Nilsson, J., Paolo, F. S., et al. (2020). Pervasive ice sheet mass loss reflects competing ocean and atmospheric processes. *Science*, 368(6496), 1239–1242. https://doi.org/10.1126/science.aaz5845
- Steger, C. R., Reijmer, C. H., van den Broeke, M. R., Wever, N., Forster, R. R., Koenig, L. S., et al. (2017). Firn meltwater retention on the Greenland ice sheet: A model comparison. Frontiers of Earth Science, 5. https://doi.org/10.3389/feart.2017.00003
- Thompson-Munson, M., Wever, N., Stevens, C. M., Lenaerts, J. T. M., & Medley, B. (2023). An evaluation of a physics-based firn model and a semi-empirical firn model across the Greenland Ice Sheet (1980–2020). *The Cryosphere*, 17(5), 2185–2209. https://doi.org/10.5194/ tc-17-2185-2023
- Torinesi, O., Fily, M., & Genthon, C. (2003). Variability and trends of the summer melt period of Antarctic ice margins since 1980 from microwave sensors. *Journal of Climate*, 16(7), 1047–1060. https://doi.org/10.1175/1520-0442(2003)016<1047:VATOTS>2.0.CO;2
- Trusel, L. D., Frey, K. E., Das, S. B., Karnauskas, K. B., Kuipers Munneke, P., van Meijgaard, E., & van den Broeke, M. R. (2015). Divergent trajectories of Antarctic surface melt under two twenty-first-century climate scenarios. *Nature Geoscience*, 8(12), 927–932. https://doi. org/10.1038/ngeo2563
- Vandecrux, B., Mottram, R., Langen, P. L., Fausto, R. S., Olesen, M., Stevens, C. M., et al. (2020). The firm meltwater Retention Model Intercomparison Project (RetMIP): Evaluation of nine firm models at four weather station sites on the Greenland ice sheet. *The Cryosphere*, 14(11), 3785–3810. https://doi.org/10.5194/tc-14-3785-2020
- van Wessem, J. M., van de Berg, W. J., Noël, B. P. Y., van Meijgaard, E., Amory, C., Birnbaum, G., et al. (2018). Modelling the climate and surface mass balance of polar ice sheets using RACMO2—Part 2: Antarctica (1979–2016). *The Cryosphere*, 12(4), 1479–1498. https://doi. org/10.5194/tc-12-1479-2018
- van Wessem, J. M., van de Berg, W. J., & van den Broeke, M. R. (2023). Data set: Monthly averaged RACMO2.3p2 variables (1979-2022): Antarctica [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7845736
- Weber Hoen, E., & Zebker, H. A. (2020). Penetration depths inferred from interferometric volume decorrelation observed over the Green-land Ice Sheet. IEEE Transactions on Geoscience and Remote Sensing, 38(6), 2571–2583. https://doi.org/10.1109/36.885204
- Wever, N., Fierz, C., Mitterer, C., Hirashima, H., & Lehning, M. (2014). Solving Richards Equation for snow improves snowpack meltwater runoff estimations in detailed multi-layer snowpack model. *The Cryosphere*, 8(1), 257–274. https://doi.org/10.5194/tc-8-257-2014
- Wever, N., Schmid, L., Heilig, A., Eisen, O., Fierz, C., & Lehning, M. (2015). Verification of the multi-layer SNOWPACK model with different water transport schemes. *The Cryosphere*, 9(6), 2271–2293. https://doi.org/10.5194/tc-9-2271-2015
- Wever, N., Würzer, S., Fierz, C., & Lehning, M. (2016). Simulating ice layer formation under the presence of preferential flow in layered snowpacks. *The Cryosphere*, 10(6), 2731–2744. https://doi.org/10.5194/tc-10-2731-2016
- Wille, J. D., Favier, V., Dufour, A., Gorodetskaya, I. V., Turner, J., Agosta, C., & Codron, F. (2019). West Antarctic surface melt triggered by atmospheric rivers. *Nature Geoscience*, 12(11), 911–916. https://doi.org/10.1038/s41561-019-0460-1
- Zwally, H. J. (1977). Microwave emissivity and accumulation rate of polar firn. Journal of Glaciology, 18(79), 195–215. https://doi.org/10.1017/s0022143000021304
- Zwally, H. J., & Fiegles, S. (1994). Extent and duration of Antarctic surface melting. Journal of Glaciology, 40(136), 463–476. https://doi. org/10.3189/s0022143000012338